Author's response to reviews

Title: Deletion Xq27.3q28 that includes IDS and FMR1 in female patient with global developmental delays and nonrandom X inactivation

Authors:

Lauren S Marshall (lsmarsha@uci.edu)
Julie Simon (Julie.Simon@multicare.org)
Tim Wood (tim@ggc.org)
Mei Peng (Mei.X.Peng@questdiagnostics.com)
Renius Owen (Renius.X.Owen@questdiagnostics.com)
Gary S Feldman (GFeldman@memorialcare.org)
Michael V Zaragoza (mzaragoz@uci.edu)

Version: 2 Date: 11 September 2012

Author's response to reviews: see over
September 11, 2012

Tim Sands, Ph.D.
Executive Editor, BioMed Central

Re: Revision of MS: 1711317179495498

Dear Dr. Sands,

I would like to submit for your consideration the case report, “Deletion Xq27.3q28 that includes IDS and FMR1 in female patient with global developmental delays and nonrandom X inactivation,” by Lauren S. Marshall, Julie Simon, Tim Wood, Mei Peng, Renius Owen, Gary S. Feldman, and Michael V. Zaragoza.

As advised, we revised the manuscript to confirm in the consent statement that the parents gave permission for publication of the case, their own clinical details and for the publication of the images (Consent section, page 11).

Thank you for your consideration, Dr. Sands

Sincerely,

Michael V. Zaragoza, M.D., Ph.D.
Assistant Professor,
Department of Pediatrics, Division of Genetics & Metabolism, and
Department of Biological Chemistry
The University of California, Irvine, School of Medicine
2042 Hewitt Hall, Irvine, CA 92697-3940
lab office 00-1-949-824-8813; clinical office 00-1-714-456-5791
Email: mzaragoz@uci.edu